Author’s response to reviews

Title: Study protocol of cost-effectiveness and cost-utility of a biopsychosocial multidisciplinary intervention in the evolution of non-specific sub-acute low back pain in the working population: cluster randomised trial

Authors:

Anna Berenguera (aberenguera@idiapjgol.org)
Enriqueta Pujol-Ribera (epujol@idiapjgol.org)
Teresa Rodriguez-Blanco (trodriquez@idiapjgol.org)
Concepció Violan (cviolan@idiapjgol.org)
Marc Casajuana (mcasajuana@idiapjgol.org)
Nelleke DeKort (dekortnelleke@gmail.com)
Marta Trapero-Bertran (marta.trapero-bertran@brunel.ac.uk)

Version: 6 Date: 5 August 2011

Author’s response to reviews:

Dear Editor,

We resubmit our manuscript entitled “The cost-effectiveness and cost-utility of a biopsychosocial multidisciplinary intervention in non-specific sub-acute low back pain in the working population in Catalonia” for publication in the BMC Musculoskeletal disorders.

We have achieved the current clinical trials registration: ISRCTN58719694

Thanks,

Dr. Enriqueta Pujol-Ribera